Author's response to reviews

Title: Antioxidant activity of rosemary (Rosmarinus officinalis L.) essential oil and its hepatoprotective potential

Authors:

Aleksandar Raskovic (araskovic@hotmail.com)
Isidora Milanovic (lola973@yahoo.com)
Nebojsa Pavlovic (nebojsa.pavlovic@gmail.com)
Tatjana Cebovic (cebovictatjana@gmail.com)
Sasa Vukmirovic (sasavukmirovic99@gmail.com)
Momir Mikov (momir.mikov@otago.ac.nz)

Version: 3 Date: 22 May 2014

Author's response to reviews: see over
Dear Reviewers,

On behalf of the co-authors, I would like to thank you for helping us with your comments and suggestions to improve our initial version of the manuscript and for accepting all changes we have made after first revision.

Reviewer #3

Dear Dr. Jeong,

Thank you for your precise comments. We agreed that all your suggestions are essential and accepted them. These changes are highlighted in the manuscript.

1. Reviewer’s comment: On page 11, the last paragraph "Antioxidant activity of rosemary essential oils was reported to be derived partly from the presence of phenolic groups" may need proper reference(s).


2. Reviewer’s comment: High dose of alpha-pinene might have a toxic effect too.


3. Reviewer’s comment: On page 16, first line, "a small amount can cause poisoning in small amounts in children" should be rephrased to "a small amount can cause poisoning in children"

Our action: We corrected this typographical error, which is highlighted in the text.

Our team hopes that we have managed to properly consider and act upon the comments and suggestions raised.

Yours sincerely,

Nebojša Pavlović, corresponding author